
 
\documentclass[11pt]{report}
 
\usepackage[margin=0.5in]{geometry} 
\usepackage{amsmath,amsthm,amssymb,verbatim}


 
\begin{document}

\large{2015 Spring Math 5043 \quad Assignment 2. \qquad Due: Feb 13, 2015}
\medskip

%\large{Name: }
\bigskip

1. Hatcher Exercise 1.1.8

\medskip


2. Hatcher Exercise 1.1.9

\medskip

3. Hatcher Exercise 1.1.10

\smallskip

Hint: First do this for $T^2$. Use the presentation of the torus as a rectangle with opposite sides identified.

\medskip

4. Hatcher Exercise 1.2.9

\medskip

5. Hatcher Exercise 1.2.11

\medskip

6. Hatcher Exercise 1.2.16

\bigskip

Fruit for thought, not to be graded:

\smallskip

I. Hatcher Exercise 1.2.22 (Wirtinger presentation)



\end{document}




















